Author's response to reviews

Title: Persons with dementia missing in the community: Is it wandering or something unique?

Authors:

Meredeth A Rowe (mrowe@ufl.edu)
Sydney S Vandeveer (sydneyvand@gmail.com)
Greenblum A Catherine (cgreenblum@aol.com)
Cassandra N List (cnlist23@gmail.com)
Rachael M Fernandez (rachael.fernandez@me.com)
Natalie E Mixson (nemixson@gmail.com)
Hyo C Ahn (hcahn@ufl.edu)

Version: 4 Date: 2 June 2011

Author's response to reviews:

Authors' contributions were reformatted to a paragraph and a sentence was added that all authors read the final version.